Author’s response to reviews

Title: A grounded theory of regaining normalcy and reintegration of women with obstetric fistula in Kenya.

Authors:
Anne Khisa (annekhisa@gmail.com)
Isaac Nyamongo (inyamongo@uonbi.ac.ke)
Grace Omoni (omonigrace@hotmail.com)
Rachel Spitzer (rfspitzer@sympatico.ca)

Version: 1 Date: 06 Jul 2018

Author’s response to reviews:

Dear sir/ Madam,

We appreciate the substantive commentary on our manuscript entitled ‘A grounded theory of regaining normalcy and reintegration of women with obstetric fistula in Kenya’. We have taken all points raised and resulted in a hopefully much more improved manuscript.

Herein enclosed are our responses and corrections on the manuscript based on the reviewers’ comments in a word document titled ‘Response to reviewers comments’, attached. Throughout the article, text has been edited for better clarity and concision. Reference numbers have also been changed to reflect the new citations, and reordering of sections. Due to the recommended changes, our manuscript word count has increased to 9446 words.

We are once again thank you and hope to hear from you regarding our revised manuscript.